\documentclass[12pt]{book}

\input{../../header}

\renewcommand{\exercisesolutions}{here}  % 0: here, 1: chapter, 2: end

\lstset{inputpath=../code}
\graphicspath{{figures/}}

\typein[\pagenumber]{Number of first page}
\typein[\chapternumber]{Chapter number}
\setcounter{page}{\pagenumber}
\setcounter{chapter}{\chapternumber}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{document} 

\include{simulations}

\section{TODO}
\begin{itemize}
\item draw (and plot) random numbers
\item draw (and plot) random functions
\item euler forward, odeint
\item introduce derivatives which are also needed for fitting (move box from there here)
\end{itemize}

\end{document}
